Author's response to reviews

Title: Internet-based Self-Assessment after the Tsunami: lessons learned

Authors:

Stefan Vetter (stefan.vetter@access.uzh.ch)  
Astrid Rossegger (astrid.rossegger@me.com)  
Thomas Elbert (thomas.elbert@uni-konstanz.de)  
Juliane Gerth (juliane.gerth@gmail.com)  
Frank Urbaniok (frank.urbaniok@ji.zh.ch)  
Arja Laubacher (arja.laubacher@ji.zh.ch)  
Wulf Rossler (roessler@dgsp.uzh.ch)  
Jérôme Endrass (jerome.endrass@me.com)

Version: 5 Date: 30 July 2010

Author's response to reviews:

Dear reviewers, dear editors,

Three persons have reviewed the manuscript in the first instance. We took up all of their suggestions and in consequence spent a lot of time and effort in re-writing and revising the manuscript. Two of the reviewers came to the conclusion that the revision was successful. The third reviewer, Peter van der Velden, has asked in his second review to make fundamental changes such as shifting the focus of the paper or the approach of data analysis. We appreciate his efforts in reviewing the manuscript as well as all the input we have received from him. However, the second round of the review process, seems not to be the appropriate time to take up fundamental changes to the focus or the analysis of a manuscript. We have therefore taken up all the suggestions of Peter van der Velden, which did not concern fundamental changes. We hope that this meets with the editors approval.